Author’s response to reviews

Title: Incidence and effects of Varicella Zoster Virus infection on academic activities of medical undergraduates - a five-year follow-up study from Sri Lanka

Authors:

Suneth B Agampodi (sunethagampodi@yahoo.com)
Samath D Dharmaratne (samathd@pdn.ac.lk)
Vasanthi Thevanesam (vasanthishevanesam@yahoo.com)
Sameera Dassanayake (sdasanayake1981@yahoo.com)
Prabhashini Kumarihami (prabhashinikumarihamy@yahoo.com)
Ashani Rathnayaka (aashanirm@yahoo.com)

Version: 4 Date: 10 May 2010

Author’s response to reviews:

Dear Editors,

I have done the corrected revisions to suit the manuscript according to the journal style. I have changed the title in to sentence case, re-uploaded the edited figures and removed track changes.

Please contact me if you need further clarification regarding the manuscript.

best wishes

Suneth